\section{Formatierung}
% -- ------------------------------------------------------------------------------------------ --
% -- Rechtliche Regularien in deutschland
% -- ------------------------------------------------------------------------------------------ --
\subsection{Subsection}
% -- ------------- --
% -- Subsubsection --
% -- ------------- --
\subsubsection{Subsubsection}
\label{sec:Subsubsection}

\ref{sec:Subsubsection}

\ref{fig:Abb01}

\pageref{fig:Abb01}

\ac{AG}

Lorem ipsum
\\

\cite{Q101}

\parencite{Q201}

(\cite*[vgl. §20 Abs.1][]{Q301})

\cite*[§20 Abs.1][]{Q301}

\begin{figure}[h]
    \centering
    \caption{\textit{Überschrift}}
    \includegraphics[width=\textwidth]{Abb01_Rechtspyramide.png} \\
    In Anlehnugn an \cite{Q101}
    \label{fig:Abb01}
\end{figure}